Author’s response to reviews

Title: Osteoporosis in Canadian Adult Cystic Fibrosis Patients: A Descriptive Study

Authors:

Ms Christine Brenckmann (cbrenckmann@hotmail.com)
Dr Alexandra Papaioannou (papaioannou@hhsc.ca)
Andy Freitag (freitaga@mcmaster.ca)
Rosamund Hennessey (hennessey@hhsc.ca)
Suzanne Hansen (shansen@hhsc.ca)
George Ioannidis (g.ioannidis@sympatico.ca)
Colin Webber (webberc@mcmaster.ca)
Jonathan Adachi (jd.adachi@sympatico.ca)

Version: 4 Date: 20 Jun 2003

Comments from Dr. Larry Lands
- added number of patients receiving oral and intravenous corticosteroids, with mean [SD] dose and range
- revised lower limit of normal for serum vitamin D3 to 25 rather than 16

Dr. Robert Aris
- commented that changes in BMD related to use of etidronate plus calcium were not statistically significant
- commented on sub-group analysis in discussion